Author's response to reviews

Title: Are behavioural interventions effective in increasing physical activity at 12-36 months in adults aged 55 to 70 years? A systematic review and meta-analysis

Authors:

Nicola Hobbs (nicki.hobbs@ncl.ac.uk)
Alan Godfrey (alan.godfrey@ncl.ac.uk)
Jose Lara (jose.lara@ncl.ac.uk)
Linda Errington (linda.errington@ncl.ac.uk)
Thomas D Meyer (thomas.meyer@ncl.ac.uk)
Lynn Rochester (lynn.rochester@ncl.ac.uk)
Martin White (martin.white@ncl.ac.uk)
John C Mathers (john.mathers@ncl.ac.uk)
Falko F Sniehotta (falko.sniehotta@ncl.ac.uk)

Version: 4 Date: 22 February 2013

Author's response to reviews: see over
Subject: Editorial Change to Revised Manuscript
22nd February 2013

Dear Editor Barnard,

Thank you for email dated 21st February 2013 asking us to make one further editorial change to our manuscript entitled “Are behavioural interventions effective in increasing physical activity at 12-36 months in adults aged 55 to 70 years? A systematic review and meta-analysis”.

As requested we have now given the role of all authors in the Authors’ contribution section of the manuscript.

We are very pleased that reviewer 2 was satisfied with our responses and would like to thank the reviewers again for providing helpful comments.

Yours sincerely,

Nicola Hobbs, PhD
CPsychol, Health Psychologist
Institute of Health & Society
Baddiley-Clark Building
Richardson Road
Newcastle University
Newcastle upon Tyne
NE2 4AX

Email: nicki.hobbs@ncl.ac.uk
Tel: +44 (0)191 222 8915
Fax: +44 (0)191 222 6043